References

L. Dornhoff, Group representation theory.

C. Curtis and I. Reiner, Methods of representation Theory. W. Fulton, Young Tableaux.

W. Fulton and J. Harris, Representation Theory a first course. J. Green, Polynomial representations of GL_n.

Brian Hall, Lie Groups, Lie Algebras, and Representations An elementary introduction. G. James and A Kerber, The representation Theory of the symmetric algebra. A. Knapp, Representation Theory of Semisimple Groups. I. G. Macdonald, Symmetric Functions and Hall Polynomials. S. Martin, Schur algebras and representation theory.

T. Neumann, Representations of the General linear group, Google “Timo Neumann Representations of the general linear group”.

B. Sagan, The symmetric Group Representations, Combinatorial Algorithms, and Symmetric functions.